

Author's response to reviews

Title: Antitumour responses induced by a cell-based Reovirus vaccine in murine lung and melanoma models

Authors:

Ciorsdan Champion (ciorsdan.champion@ucc.ie)

Declan Soden (d.soden@ucc.ie)

Patrick Forde (p.forde@ucc.ie)

Version: 3 Date: 28 Jun 2016

Author's response to reviews:

1. Please move the 'Methods' section and place it inbetween the "Background" and "Results" sections, as per our Submission guidelines (<http://bmccancer.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article>)

Methods section has been moved as requested

2. Thank you for providing a statement in the "Availability of data and materials" section. However, submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. (see

<http://www.biomedcentral.com/getpublished/editorial-policies#DataandMaterialRelease>)

Please detail where the data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested, please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the 'Availability of data and Materials' section) that data will not be shared, and state the reason.

Section updated.